Author's response to reviews

Title: A Case Report: Familial Glucocorticoid Deficiency associated with Familial Focal Segmental Glomerulosclerosis.

Authors:

NANIK RAM (nanik.ram@aku.edu)
ALI ASGHAR (ali.asghar@aku.edu)
NAJMUL ISLAM (najmul.islam@aku.edu)

Version: 2 Date: 25 October 2012

Author's response to reviews: see over
To The Editor

BMC Endocrine Disorders Journal

Subject: **Re-Submission of case report**

Dear Sir/Madam

Following are the corrections have been made according to reviewers report.

**Reviewer: Karen Heath**

1. Unfortunately molecular analysis for genetic mutation is not possible due to non availability of test in our country and through research collaboration.

2. In discussion with reference to Meimaridou et al article description regarding genetic mutations involved in FGD has been added.

3. Suggested editing of English phrases has been done.

4. References editing has been done according to BMC endocrine disorder journal.

**Reviewer: Serap Turan**

1. Addison’s disease has been replaced with primary adrenal insufficiency.

2. Pedigree chart has been added.

3. In discussion Metwalley KAK et al case is compared with our case with regard to clinical features. Correction has been made.

4. Etiology of adrenal failure of our case has been discussed. Genetic link and underlying factors/facilitators/mechanism between these two conditions not yet known. Prevalence of FSGS in adrenal failure in not known, there are only case reports.
Reviewer: Nils Krone

1. Pedigree diagram included.
2. Paper checked and corrected for grammar.
3. Reported association is less likely to be coincidence since 2 of 3 siblings have both conditions.
4. In introduction, word association been replaced with reported cases of FSGS and other endocrine disorders.
5. In case presentation, change of diagnosis based on lab results obtained off medicine. It has been clarified.
6. Discussion brought in view of literature by discussing and comparing our case with other reported cases in literature.

Yours Sincerely

Nanik Ram